Author’s response to reviews

Title: Calf circumference as an independent indicator to insulin resistance and non-alcohol fatty liver disease: the REACTION study

Authors:

Weiwei Zhang (amanda_zww@126.com)
Zhen Yang (yangzhen1020@hotmail.com)
Yixin Niu (niuniu@126.com)
Xiaoyong Li (lixiaoyong@126.com)
Lingfei Zhu (zhuzhu@126.com)
Shuai Lu (lushuai@126.com)
Hongmei Zhang (zanghongmei@126.com)
Jiangao Fan (fanjiangao@126.com)
Guang Ning (ningguang@163.com)
Li Qin (qinli@163.com)
Qing Su (suqingxinhua@163.com)

Version: 3 Date: 04 Apr 2017

Author’s response to reviews:

Dear Editor:

Our manuscript has been extensively revised in accordance with the editors’ comments.

Minor Comments:

1. Unfortunately, throughout the manuscript there is still significant text overlap with other publications. While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. We would therefore be grateful if you could provide a
justification for the overlap in text between your manuscript and other sources. Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated. The report showing text overlap has been attached to this email to further help you with your revisions.

Answers:

We thank for the editors’ reminds. According to editors’ suggestion, we have revised the part, which is same with the previous publications, in the manuscript.

2. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Answers:

We are sincerely appreciative of editors’ comment. According to the editors’ advice, we have add the email addresses for all authors on the title page.

3. Please upload your Tables as separate table files rather than within the same file.

Answers:

We thank for the editors’ instruction. We have uploaded my Tables as separate table files.

4. We would also like to ask for you to provide more justification for the contributions of XYL, LFZ, SL, HMZ, JGF, GN, as currently they do not automatically qualify for authorship. Contribution of reagents/materials/analysis tools, alone, does not usually justify authorship.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND
c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Answers:

We thank for the editors’ suggestions. We have provided more justification for the contributions of XYL, LFZ, SL, HMZ, JGF, GN in the part of Authors’ contributions.